Aalborg Universitet



Hip fracture risk in patients with alcoholic cirrhosis

a population-based study using English and Danish data

Otete, H; Deleuran, T; Fleming, K M; Card, T R; Aithal, G P; Jepsen, P; West, J

Published in: Journal of Hepatology

DOI (link to publication from Publisher): 10.1016/j.jhep.2018.04.002

Creative Commons License CC BY-NC-ND 4.0

Publication date: 2018

Document Version Accepted author manuscript, peer reviewed version

Link to publication from Aalborg University

Citation for published version (APA):

Otete, H., Deleuran, T., Fleming, K. M., Card, T. R., Aithal, G. P., Jepsen, P., & West, J. (2018). Hip fracture risk in patients with alcoholic cirrhosis: a population-based study using English and Danish data. *Journal of Hepatology*, *69*(3), 697-704. https://doi.org/10.1016/j.jhep.2018.04.002

General rights

Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights.

- Users may download and print one copy of any publication from the public portal for the purpose of private study or research.
- You may not further distribute the material or use it for any profit-making activity or commercial gain You may freely distribute the URL identifying the publication in the public portal -

Take down policy

If you believe that this document breaches copyright please contact us at vbn@aub.aau.dk providing details, and we will remove access to the work immediately and investigate your claim.

Accepted Manuscript

Hip fracture risk in patients with alcoholic cirrhosis: a population-based study using English and Danish data

H. Otete, T. Deleuran, K.M. Fleming, T.R. Card, G.P. Aithal, P. Jepsen, J. West

PII:	\$0168-8278(18)31978-0
DOI:	https://doi.org/10.1016/j.jhep.2018.04.002
Reference:	JHEPAT 6932
To appear in:	Journal of Hepatology
Received Date:	29 September 2017
Revised Date:	4 April 2018
Accepted Date:	6 April 2018



Please cite this article as: Otete, H., Deleuran, T., Fleming, K.M., Card, T.R., Aithal, G.P., Jepsen, P., West, J., Hip fracture risk in patients with alcoholic cirrhosis: a population-based study using English and Danish data, *Journal of Hepatology* (2018), doi: https://doi.org/10.1016/j.jhep.2018.04.002

This is a PDF file of an unedited manuscript that has been accepted for publication. As a service to our customers we are providing this early version of the manuscript. The manuscript will undergo copyediting, typesetting, and review of the resulting proof before it is published in its final form. Please note that during the production process errors may be discovered which could affect the content, and all legal disclaimers that apply to the journal pertain.

Hip fracture risk in patients with alcoholic

cirrhosis: a population-based study using

English and Danish data.

Otete H, Assistant professor^{1,2}, **Deleuran T**, Post Doc^{3,4}, **Fleming KM**, Senior Lecturer⁵, **Card TR**, Clinical Associate Professor^{1,6}, **Aithal GP**, Professor^{6,7}, **Jepsen P**, Clinical Associate Professor^{3,8}, **West J**, Professor^{1,6}.

Affiliations

¹ Division of Epidemiology and Public health, University of Nottingham, City Hospital Campus, Nottingham, NG5 1PB, United Kingdom.

² School of Pharmacy, University of Nottingham, Nottingham, NG7 2RD, United Kingdom.

³ Department of Hepatology and Gastroenterology, Aarhus University Hospital, Aarhus, Denmark.

⁴Department of Gastroenterology and Hepatology, Aalborg University Hospital, Denmark

⁵ Department of Public health and Policy, University of Liverpool, Liverpool, L69 3GB United Kingdom.

⁶ Nottingham Digestive Diseases Centre, School of Medicine, University of Nottingham, Nottingham, Uk

⁷NIHR Nottingham Biomedical Research Centre at the Nottingham University Hospitals NHS Trust and University of Nottingham, Nottingham, UK.

⁸ Department of Clinical Epidemiology, Aarhus University Hospital, Denmark

Corresponding author: Harmony Otete, <u>harmony.otete@nottingham.ac.uk</u>, C03 School of Pharmacy, University Park, University of Nottingham, NG7 2RD, United Kingdom.

Word count:3,974

Disclosures: All authors have completed the ICMJE uniform disclosure form at <u>http://www.icmje.org/coi_disclosure.pdf</u>. HO, JW, PJ, KMF,TDE and TC have nothing to disclose. AGP reports other from Aegerion, other from Shire, other from A star D3, other from Agios, outside the submitted work.

Contributors

JW, KMF, and PJ conceived the study and planned it with HO and TD. HO and TD carried out the analysis. HO, TD, KMF, TC, AGP, PJ and JW contributed to the interpretation of data. OH, JW, and KMF drafted the article. HO, TD, KMF, TC, AGP, PJ and JW revised the article for important intellectual content. All authors gave final approval of the version to be published. JW is the guarantor.

Key words: Hip fracture, alcoholic liver disease, cirrhosis, mortality, epidemiology.

What is already known about the subject?

- Cirrhosis is on the rise with little proven treatment that affects its natural history or mortality
- The risk of hip fracture is increased among some groups of people with cirrhosis
- Hip fracture confers a large morbidity and mortality effect immediately following the event
- There is lack of evidence on the absolute and relative hip fracture risk specifically in patients with alcoholic cirrhosis, taking into account competing risk of death.

What are the new findings?

- Hip fracture risk is raised by at least 5 fold in an alcoholic cirrhosis population compared to population controls of similar age and sex
- The increased risks of hip fracture is higher in younger adults
- The cumulative incidence of hip fracture (having taken account of the competing risk of death) is greater than that of hepatocellular carcinoma (HCC) in this population.

How might it impact on clinical practice in the foreseeable future?

Hip fracture is, via several interventions, potentially preventable. Given that the
reported risk of hip fracture in people with cirrhosis is higher than that of HCC for
which there is national surveillance, our findings support the need for more effort
towards fracture prevention in this population as it could benefit both the individual
and society burden.

Downloaded for Anonymous User (n/a) at Aalborg University Hospital from ClinicalKey.com by Elsevier on April 23, 2018. For personal use only. No other uses without permission. Copyright ©2018. Elsevier Inc. All rights reserved.

Abstract

Background & aims: Cirrhosis is a risk factor for osteoporosis and fractures. However, little is known of the actual risk of hip fractures in patients with alcoholic cirrhosis. Using linked primary and secondary care data from the English and Danish nationwide registries, we quantified the hip fracture risk in two national cohorts of patients with alcoholic cirrhosis.

Methods: We followed 3,706 English and 17,779 Danish patients with a diagnosis of alcoholic cirrhosis, and we identified matched controls from the general populations. We estimated hazard ratios (HR) of hip fracture for patients versus controls, adjusted for age, sex and comorbidity.

Results: The 5-year hip fracture risk was raised both in England (2.9% vs 0.8% for controls) and Denmark (4.6% vs 0.9% for controls). With confounder adjustment, patients with cirrhosis had 5-fold (adjusted HR 5.5 (95% CI 4.3-6.9)), and 10-fold (adjusted HR 9.9 (95% CI 8.9-11.0)) increased rates. This association between alcoholic cirrhosis and risk of hip fracture showed significant interaction with age (p<0.001), being stronger in younger age groups (under 45 years HR: 17.9 and 16.6 respectively for English, Danish) than in patients over 75 years (HR 2.1 and 2.9 respectively). In patients with alcoholic cirrhosis, 30-day mortality following a hip fracture was 11.1% in England and 10.0% in Denmark, giving age-adjusted post-fracture mortality rate ratios of 2.8(95% CI 1.9-3.9) and 2.0(95% CI 1.5-2.7), respectively.

Conclusions: Patients with alcoholic cirrhosis have a markedly increased risk of hip fracture and post-hip fracture mortality compared with the general population.

Lay summary

Alcoholic cirrhosis creates a large public health burden and is a risk factor for bone fractures. Based on data from England and Denmark, we found that hip fractures occur more than 5 times more frequently in people with alcoholic cirrhosis than in people without the disease. Additionally, the aftermath of the hip fracture is severe such that Up to 11% of patients with alcoholic cirrhosis die

within 30 days after their hip fracture. These results suggest that efforts directed towards fracture prevention in people with alcoholic cirrhosis could be beneficial.

SCRIF

Keywords: cirrhosis; epidemiology; fracture

Introduction

An important and often studied complication of chronic liver disease is osteoporosis (1–5). Osteoporosis may be asymptomatic, but increases the risk that minor accidents result in bone fractures, particularly of the distal radius and proximal femur (1,2). Hip fractures in particular have a significant impact on health, productivity and life expectancy (6,7), and the 30-day mortality is estimated to be up to 10% (8,9). In patients specifically with alcoholic cirrhosis, the risk of these fractures may be further increased due to the direct effect of high levels of alcohol use (10), or minimal hepatic encephalopathy (11). However, due to a dearth of studies that have actually quantified the incidence of fractures in chronic liver disease, the absolute risk of hip fractures in people with alcoholic cirrhosis is unknown.

Available estimates on the occurence of fractures in hepatic disorders originate from small studies on hetergenous populations that were carried out almost 10 years ago. Only two recent studies have addressed the risk of fractures in alcoholic liver disease specifically (5,12), but – like previous studies (13,14) – they studied relative risks, not absolute risks and did not do a comprehensive analysis by fracture type, so estimates of hip fracture specifically are unavailable. Clinical practitioners and policy makers would be better placed to inform and guide patients regarding prevention strategies if their actual risk of serious injury, such as hip fracture, was known. Indeed, while the relative risk of a hip fracture is useful in terms of identifying a group in which targeted intervention could be valuable, without absolute risks it is impossible to judge the risks and benefits of potential interventions. In a cirrhosis

population it is crucial also when quantifying risk to take account of the fact that there is a high mortality (15,16). Only by accounting for this competing risk and demonstrating the cumulative risk of hip fracture can we accurately quantify the problem. We therefore have performed a large study using two national cohorts in England and Denmark to examine the absolute and relative risk of incident hip fracture in patients with alcoholic cirrhosis compared with the general population, and to compare and contrast our findings given the differences in the populations of the two countries.

Methods

Data sources

English data was obtained from the linked Clinical Practice Research Datalink (CPRD) and Hospital Episodes Statistics (HES) database (17,18), while Danish data was obtained from The Danish National Patient Registry (NPR) (19) and the Danish civil registration system (20).

Clinical Practice Research Datalink (CPRD) and Hospital Episodes Statistics (HES)

The CPRD consists of computerised general practice records for over 15 million patients in the UK of which 4.4 million are active, alive and currently registered. With a coverage of 4.4 million, it covers approximately 6% of the UK population. Recorded data includes patient demography, diagnoses and test results entered during general practice appointments or following communication from secondary care. These data are coded electronically using the hierarchical Read coding system (21) and are subjected to regular data quality checks and audits to ensure 95% inclusion of prescribing and morbidity events. Since 1997, the CPRD has an established linkage to the HES database, which contains information on all admissions to the state-funded National Health Service (NHS) hospitals in England; including secondary care diagnoses coded using the International Classification of Disease (ICD) edition 10 (ICD-10). Approval for use of this linked data for this study was given by the Scientific and Ethical Committee of the CPRD (15_073RAR).

National Patient Registry (NPR)

The Danish NPR is a nationwide registry that holds data on all admissions to non-psychiatric hospitals since 1977 and from outpatient and emergency room visits since 1995 in Denmark. The data includes dates and diagnoses coded in accordance with the ICD-10 from 1994 and ICD-8 before that. The Danish Central Office of Civil Registration (CPR) monitors the vital

status of all Danish citizens (population 5.6 million) continuously and issues a unique personal identifier to everyone at birth or migration to Denmark. This number enables individual-level linkage with the NPR.

Study population

All adults (>18 years) with a first time diagnosis of alcoholic cirrhosis between 1997 and 2014 were identified. Alcoholic cirrhosis was defined by the presence of a diagnostic code (Read or ICD-10 code) for alcoholic cirrhosis. Our code lists were adapted from previously validated definitions (22,23). We defined the 'index date' of each patient as the first date of a recorded diagnostic code for alcoholic cirrhosis. Among the patients identified within the English data, we excluded those with an index date within 1 year of registration with a general practice in order to avoid prevalent cases. For each included patient, we identified up to 10 general population controls frequency-matched on sex, age (within 5 years) and by registration at the same general practice, (general practice matching was for patients identified in England only). The matched controls had to be alive and without a diagnostic or therapeutic code for cirrhosis, oesophageal varices and/or portal hypertension within their own general practice or hospital record.

Controls identified for the English cohort were assigned an index date, which was a randomly generated date from 1 year after the start of the linked dataset (1997) up to the date they left their general practice or died. For the Danish cohort, controls were assigned the same index date as the cirrhosis patient they were matched to. Patients with a diagnosis of a hip fracture (as defined below) before the index date were excluded from both cohorts.

Study period

Each patient and control was followed from the index date to a diagnosis for a hip fracture, date of death, date of moving out of their general practice (for English data only), or end of follow-up (31/12/2014).

Study definitions

Primary outcome

The primary outcome was time to first diagnosis for any hip fracture (ICD-10 S720-S729; Read codes 820 A, 820 B, 8210, S30..00, S30..11, S302.00, S30y.00, S30y.11, S30z.00) as recorded on the general practice, hospital, or patient registry records (24).

Severity of cirrhosis

Patients with alcoholic cirrhosis were classified as being in a compensated or decompensated disease state according to the Baveno clinical staging classification that takes into account the absence or presence of variceal bleed, and ascites (but specifically does not include hepatic encephalopathy) (25) . Variceal bleed or ascites was defined if there was a relevant diagnostic code (ICD-10 code: R18, I850, K920, K921, K922, I859, I864, I982) or procedure code (OPCS4 code: T461, T462, G104, G108, G109, G144, G174, G437) up to 1 year before the date of diagnosis of alcoholic cirrhosis; cirrhosis patients without such codes were defined as having compensated disease. We specifically identified evidence of hepatic encephalopathy by searching patient records for relevant diagnostic codes (ICD 10: E51.2, G31.2, K70.4, K72.1).

Other cirrhosis aetiologies

We reported whether our patients with alcoholic cirrhosis also had cirrhosis of other etiology including viral hepatitis, autoimmune liver disease (primary biliary cholangitis, primary sclerosing cholangitis and autoimmune hepatitis), haemochromatosis, Wilson's disease and alpha-1 antitrypsin deficiency at any time prior to alcoholic cirrhosis diagnosis using methods described previously(22).

Comorbidity

For each cirrhosis patient and population control, we used secondary care data recorded within two years before the index date to compute their Charlson Comorbidity Index (CCI)

(26). The CCI is a weighted comorbidity score that is defined for use in the ICD-10. It includes myocardial infarction, congestive heart failure, peripheral vascular disease, cerebrovascular disease, mild liver disease, dementia, chronic pulmonary disease, rheumatic disease, peptic ulcer, diabetes with or without complications, hemiplegia or paraplegia, renal disease, moderate or severe liver disease, cancer, metastatic cancer, and HIV/AIDS. Diagnoses related to liver disease were excluded when calculating the CCI for this study. For clarity of presentation, we summarized the CCI in three categories: no comorbidity (CCI = 0), moderate comorbidity (CCI = 1), and multiple or severe comorbidity (CCI \geq 2).

Potential confounders

We extracted data on patients with an osteoporosis diagnosis. In addition, where available, we extracted data on drug exposure to oral and injected steroids, antidepressants, opiates and bisphosphonates. These drugs were chosen because they have been previously shown to be associated with fracture or are used as a treatment for osteoporosis(27). A density parameter for each subject was defined as the number of prescriptions received (with a maximum of 1 prescription per day per drug category permitted) per follow-up time as defined below. This prescription density was then categorised, for each drug category, into none, some and many of the specific prescriptions. The categorisation of some and many was made, for each drug, as 'less than or equal to' or 'greater than' the median density of prescriptions (having excluded those with zero prescriptions).

Statistical analyses

We calculated the rate of first hip fracture in alcoholic cirrhosis patients and population controls. Using Cox regression, we estimated the hazard ratio (HR) for a hip fracture in the alcoholic cirrhosis cohort compared with the control population. All standard Cox regression models were *a priori* adjusted for age, sex and comorbidity at baseline (Charlson comorbidity

index, 0,1 or 2). We computed HRs for subgroups defined by age group, sex and cirrhosis severity (compensated or decompensated cirrhosis) and examined effect modification by age with the likehood ratio test and graphically using a Schoenfeld residuals plot. We computed the 1- and 5- year cumulative risk of a hip fracture after alcoholic cirrhosis diagnosis using the cumulative incidence function with death without hip fracture as the competing risk (using age as the timescale). Proportional hazards assumptions were confirmed using Schoenfeld residuals and log-log plots. Finally, we identified all deaths occuring within 30 days after hip fracture diagnosis and estimated 30-day mortality and mortality risk for cirrhosis patients versus the control population.

Additional analyses

We repeated our analyses on disease severity to include patients who had a clinical diagnosis of hepatic encephalopathy in the decompensated group. This enabled us to examine the effect of hepatic encephalopathy on the association between disease severity and fracture rates. We also stratified fracture estimates by the presence or absence of osteoporosis and bisphosphonate use, as these are on the causal pathway between cirrhosis and fractures and adjusting for them in our models would have been inappropriate. We assessed the effect of drug exposure using English data where prescription records are available. Drug exposure (oral and injected steroids, antidepressants, opiates) was modelled as a categorical variable and included in the *a priori* adusted cox regression model if it conferred a 10% adjustment in the hazard ratios seen (Supplementary tables 1 and 2). Finally, to examine the role of comorbidity on fracture rates, we re-estimated the 5-year cumulative incidence of fracture in only those patients with a Charlson comorbidity score of 0 (Supplementary figure 1) i.e. no recorded comorbidity.

All statistical analyses were conducted using Stata version 13.0 (StataCorp, College Station, TX).

10

Results

Demographic and clinical characteristics

In the English data, we included 3,706 alcoholic cirrhosis patients and 36,859 matched population controls. The median age at diagnosis was 56 years, 68.0% of patients were men, and 49.8% had decompensated cirrhosis at their diagnosis. At inclusion, cirrhosis patients had more comorbidity than population controls and the prevalence of a CCI score of 1 or more was 37.7% in cirrhosis patients and 7.1% in population controls (Table 1). In the Danish data, we included 17,779 cirrhosis patients and 80,815 population controls. They were slightly older than the English cirrhosis patients (median age 57 years), but their gender composition was similar to English cirrhosis patients (66.8% were men). The prevalence of decompensated cirrhosis at the diagnosis for cirrhosis was 30% and Danish cirrhosis patients also had more comorbidity than population controls; the prevalence of a CCI score of 1 or more was 36.6% in cirrhosis patients and 5.4% in population controls (Table 1). In both England and Denmark, the majority of patients [English (88.8%) and Danish (94.6%)] had an alcohol-only cirrhosis aetiology (taking into account medical records before alcoholic cirrhosis diagnosis). Evidence of osteoporosis was found in 6.3% and 3.1% of English and Danish patients respectively.

Fracture analysis

We followed up the English population for a total of 177,717 person years, during which 485 hip fractures occurred (108 fractures in cirrhosis patients, 377 fractures in controls). The absolute hip fracture rate for patients with cirrhosis was 11.4 fractures (95% CI: 9.4-13.8) per 1000 person years but the rate among controls was much lower at 2.2 (95% CI: 2.0-2.5) per 1000 person years (Table 2). In effect, there was an excess of 9.2 fractures per 1000 person years in English cirrhosis patients. The adjusted hip fracture hazard ratio for cirrhosis patients vs. controls was 5.5 (95% CI: 4.3-6.9) (Table 2).

The corresponding numbers for the Danish population were 753,797 person years of followup and 2,491 fractures (977 fractures in cirrhosis patients and 1,514 fractures in controls). Similarly, the absolute hip fracture rate for Danish cirrhosis patients was higher than that of controls, at 16.0 (95% CI: 15.0-17.0) per 1000 person years compared to 2.2 (95% CI 2.1-2.3) per 1000 person years (Table 2). Thus, there was an excess of 13.8 fractures per 1000 person years in Danish cirrhosis patients over controls. The adjusted hip fracture hazard ratio for Danish cirrhosis patients vs. controls was 8.5 (95% CI: 7.8–9.3).

Fracture rate analysis by subgroups

In both countries the association between cirrhosis and hip fracture rate was stronger in younger age groups (aged < 45 years) than in older age (p value for interaction with age <0.001). The hazard ratio of a hip fracture in those aged 45-54 years was 14.7 (95% CI 8.6-24.9) and 13.1 (95% CI 10.9-15.7) in England and Denmark respectively, whereas for those aged 75 years and over it was 2.1 (95% CI 1.1-3.9) (England) and 3.0 (95% CI 2.1–4.2) (Denmark) (Table 2 and Figure 1). In Denmark, the association between cirrhosis and hip fracture rate was slightly stronger in men (HR 9.0, 95% CI 8.0–10.1) than in women (HR 7.9, 95% CI 6.9–9.0) but in England it was stronger in women (HR 7.1 (95% 5.0- 9.9)) than in men (HR 4.4 (95% 3.2 - 6.1)). Disease severity had no remarkable effect on absolute fracture rate in Demark, but in England, absolute fracture rates were higher in patients with decompensated disease than in compensated disease (14.2 vs. 9.4 per 1,000 person years). Including hepatic encephalopathy in the definition of disease severity had a minimal effect on the association between disease severity and fracture rates.

1-year and 5-year cumulative incidence analysis

After 1-year of follow-up, the cumulative hip fracture risk in English cirrhosis patients was 1.0% (95% CI: 0.7–1.4) compared to 0.2% (95% CI: 0.1–0.3) for population controls. The

corresponding 5-year risks were also higher for cirrhosis patients than for controls at 2.9% (95% CI 2.3-3.6) vs. 0.8% (95% CI 0.7-0.9). Likewise, in the Danish cohort, the 1-year hip fracture risk for cirrhosis patients was higher than that of controls at 1.4% (95% CI: 1.2–1.6) vs 0.2% (95% CI: 0.1–0.2). The corresponding 5-year risks were 4.6% (95% CI: 4.3–5.0) and 0.9% (95% CI: 0.8–0.9) respectively.

In both countries, the 5-year risks of hip fracture increased with increasing age and was higher for each cirrhosis age group than for the matched controls (Figure 2, Table 3). The cumulative risk among the young controls (less than 45 years) was very low, but the risk in young cirrhosis patients was remarkably high hence the increase in risk relative to the control population was greater in younger cirrhosis patients than in those over 75 years (Figure 2). We found no variation in hip fracture risk by disease severity (Table 3).

30-day mortality estimate

By 30 days post hip fracture, 11.1% (12/108) of English cirrhosis patients and 10.0% (98/977) of Danish cirrhosis patients had died compared to 5.0% (England) and 6.6% (Denmark) for the control populations (Table 4). The adjusted mortality risk ratio comparing deaths in cirrhosis patients to controls was 2.8 (95% CI: 1.9-3.9) and 2.0 (95% CI: 1.5–2.7), for England and Denmark respectively.

Additional analyses on drug use

Supplementary Table 1 provides information on drug usage for the English cirrhosis and control cohort. Patients with cirrhosis were more likely than controls to have had a prescription of opiates (42.2% vs 32.7%) and antidepresessants (14.2% vs 11.1%), but as

<u>likely to have had a prescription for oral or injected corticosteroids(13.9% vs. 14.7%)</u>. As shown in supplementary table 2, there was only minimal confounding identified following adjustment for prescription drug use.

Discussion

Using data from two national cohorts, we have shown an increased rate of hip fracture for people with alcoholic cirrhosis of between 5-fold to nearly 10-fold compared with the general population. The 5-year risk of hip fracture was 2.9% for cirrhosis patients in the England and 4.6% for patients in Denmark. In both countries, younger patients had the largest risk difference when comparing cirrhosis patients with controls. This association was due to the combination of a very low risk of hip fractures among young control patients and the contrasting high risk in young patients with alcoholic cirrhosis. We found a higher 30-day post-fracture mortality in cirrhosis patients of between 2 to 3 fold compared to population controls. Our study therefore indicates that interventions to prevent hip fracture in this population could have a significant benefit.

We chose to combine two national cohorts for this analysis so that we could compare and contrast our findings, and explore some of the differences. There were many similarities between the alcoholic cirrhosis cohorts in both countries. The age and sex profiles were very similar and the relative risks of fracture again were reasonably similar. Importantly the interaction with age showed the same pattern of relative and cumulative risks in both countries indicating its likely veracity. Our conclusions were the same irrespective of the severity of alcoholic cirrhosis suggesting that the 20% difference in the prevalence of decompensation between both cohorts was not crucial to the interpretation of our analysis.

Strengths and limitations

This is the first study to report the incidence of hip fracture in patients with alcoholic cirrhosis using prospectively collected data from two large national cohorts, with access to a suitable control population. Other strengths are its large size, competing risk adjustment, and population-based setting which means we have been able to estimate both the absolute and relative risks of hip fractures with reasonable precision and accuracy, and that these estimates are generalizable to the general populations of the two nations and possibly given the similarity between the results, to others as well. In addition, we were able to stratify estimates by age and sex, and by so doing have assessed for variation by these variables. Information on some other potential confounding factors (28,29) were either missing for some patients or absent in totality, e.g. weight bearing exercise, meaning that we could not assess their potential effect on our estimates. Our estimates were also not adjusted for alcohol use because heavy alcohol intake is an inherent part of having alcoholic cirrhosis and it would therefore be inappropriate to adjust for. Obviously, this means we do not know how much alcoholic individuals drank post-cirrhosis diagnosis nor how this might alter to their fracture risk. However, the presence of an increased risk of fracture in non-alcoholic groups with liver injury(4) argues that alcohol cannot entirely account for the relationship shown in our study.

The data sources used for this study have previously been validated and quality assured both for the diagnosis of cirrhosis (23,30) and for hip fracture (31,32). A validation of the severity of disease, as measured through decompensation, has not however been possible and as such, we may have missclassified some patients between stages of disease. Nevertheless, our previous work(33) has shown that the Baveno method of classifying decompensation accurately predicts mortality and as such appears a reasonably sound approach. Finally, while these results have important implications for fracture prevention in people with cirrhosis, this analysis does not consider the mechanism of fractures, thus it is not possible to explicitly attribute the increased risk of fractures to, for example, osteoporosis, trauma, both or some other risk factors. Although we found 6.3% (England)

15

and 3.1% (Denmark) of patients with evidence of osteoporosis, these should be interpreted with caution as recording of osteoporosis diagnosis within our data is presumed to be incomplete and therefore underestimated. <u>Future studies may consider exploring the prevalence of osteoporosis in alcoholic cirrhosis in more detail, given this lack of robust osteoporosis data in our study.</u>

Comparison with previous literature

Previous publications on the association between osteoporosis, fractures and alcholic liver disease have been with small heterogenous populations and thus far, only relative risks have been presented. In 2014, Bang et al reported a relative fracture risk in people with alcoholic cirrhosis of 2.4, but this was based on a small subgroup of patients identified within a larger cohort of patients with either chronic pancreatitis or cirrhosis (5). In another metaanalysis from the same group in 2015, based on six studies (34-39), a lower relative increase in fracture risk of 1.9 was found among people with alcoholic liver disease of mixed severity (12). Several other data for comparison originate from studies based on other chronic liver diseases. Tsai et al, estimated the risk of fractures in 3,764 patients with cirrhosis (most likely to be predominantly hepatitis-B related) compared to non-cirrhotic controls, and found a 4-fold increase in skull fracture risk in cirrhosis patients with hepatic encephalopathy (HE), and a 1.6- to 1.8-fold increase in spine, trunk and upper limb fracture risk in cirrhosis patients without HE (40). In two other studies using the same UK primary care database as our study, high relative increases in hip and wrist fracture rates of between 1.9 and 4.0 were found in patients with primary biliary cholangitis (PBC) or any chronic liver disease (4,14).

In our analysis accounting for death as a competing risk, we found no difference in the cumulative risk of fractures between patients with compensated and decompensated disease. These findings are consistent with two smaller studies on other advanced liver disease (PBC) which found no association between the occurrence of fractures and severity of liver disease (4,41).

Clinical implications and conclusion

Overall, our study has shown that hip fracture is an important clinical outcome in patients with alcoholic cirrhosis due to its frequent occurrence and severe consequence. The cumulative risk of fracture, having accounted for the competing risk of death, in these patients is approximately 1% per annum and there is more than two fold higher post fracture 30-day mortality in people with cirrhosis than in the general population. Currently, great emphasis is placed on Hepatocellular carcinoma (HCC) as an important, potentially preventable, cause of mortality in cirrhosis with several guidelines proposing national HCC surveillance to enhance detection and thus early treatment (42,43). Given that our study has shown that the cumulative incidence of hip fractures in alcoholic cirrhosis is higher than that of HCC (15,44), hip fractures may be an even more important cause of death and morbidity among people with alcoholic cirrhosis. To these facts can be added the knowledge that effective pharmacotherapy exists to treat osteoporosis and is proven to reduce hip fracture risk(45–48).

Therefore, prophylaxis to reduce hip fracture occurrence in people with alcoholic cirrhosis (48–51), may have greater potential for benefit than does the currently common practice of surveillance for HCC, and certainly warrants consideration alongside it.

References

- 1. Leslie WD, Bernstein CN, Leboff MS. AGA technical review on osteoporosis in hepatic disorders. Gastroenterology. 2003 Sep;125(3):941–66.
- Collier JD, Ninkovic M, Compston JE. Guidelines on the management of osteoporosis associated with chronic liver disease. Gut. 2002 Feb 1;50(suppl 1):i1–9.
- 3. Mounach A, Ouzzif Z, Wariaghli G, Achemlal L, Benbaghdadi I, Aouragh A, et al. Primary biliary cirrhosis and osteoporosis: a case-control study. J Bone Miner Metab. 2008;26(4):379.
- 4. Solaymani–Dodaran M, Card TR, Aithal GP, West J. Fracture Risk in People With Primary Biliary Cirrhosis: A Population-Based Cohort Study. Gastroenterology. 2006 Dec;131(6):1752–7.
- 5. Bang UC, Benfield T, Bendtsen F, Hyldstrup L, Beck Jensen J. The Risk of Fractures Among Patients With Cirrhosis or Chronic Pancreatitis. Clin Gastroenterol Hepatol. 2014 Feb;12(2):320–6.
- 6. Wolinsky FD, Fitzgerald JF, Stump TE. The effect of hip fracture on mortality, hospitalization, and functional status: a prospective study. Am J Public Health. 1997 Mar;87(3):398–403.
- 7. Bone Health and Osteoporosis: A Report of the Surgeon General. Rockville (MD): Office of the Surgeon General (US); 2004.
- 8. Sheikh HQ, Hossain FS, Aqil A, Akinbamijo B, Mushtaq V, Kapoor H. A Comprehensive Analysis of the Causes and Predictors of 30-Day Mortality Following Hip Fracture Surgery. Clin Orthop Surg. 2017 Mar;9(1):10–8.
- 9. Daugaard CL, Jørgensen HL, Riis T, Lauritzen JB, Duus BR, van der Mark S. Is mortality after hip fracture associated with surgical delay or admission during weekends and public holidays? A retrospective study of 38,020 patients. Acta Orthop. 2012 Dec;83(6):609–13.
- Zhang X, Yu Z, Yu M, Qu X. Alcohol consumption and hip fracture risk. Osteoporos Int J Establ Result Coop Eur Found Osteoporos Natl Osteoporos Found USA. 2015 Feb;26(2):531–42.
- 11. Lauridsen MM, Jepsen P, Vilstrup H. Critical flicker frequency and continuous reaction times for the diagnosis of minimal hepatic encephalopathy. A comparative study of 154 patients with liver disease. Metab Brain Dis. 2011;26(2):135.
- 12. Bang CS, Shin IS, Lee SW, Kim JB, Baik GH, Suk KT, et al. Osteoporosis and bone fractures in alcoholic liver disease: A meta-analysis. World J Gastroenterol WJG. 2015 Apr 7;21(13):4038–47.
- Tignor AS, Wu BU, Whitlock TL, Lopez R, Repas K, Banks PA, et al. High Prevalence of Low-Trauma Fracture in Chronic Pancreatitis. Am J Gastroenterol. 2010 Dec;105(12):2680–6.
- 14. Hippisley-Cox J, Coupland C. Predicting risk of osteoporotic fracture in men and women in England and Wales: prospective derivation and validation of

Downloaded for Anonymous User (n/a) at Aalborg University Hospital from ClinicalKey.com by Elsevier on April 23, 2018. For personal use only. No other uses without permission. Copyright ©2018. Elsevier Inc. All rights reserved.

QFractureScores. BMJ [Internet]. 2009 Nov 20;339. Available from: http://www.bmj.com/content/339/bmj.b4229.abstract

- 15. West J, Card TR, Aithal GP, Fleming KM. Risk of hepatocellular carcinoma among individuals with different aetiologies of cirrhosis: a population-based cohort study. Aliment Pharmacol Ther. 2017 Apr 1;45(7):983–90.
- 16. Jepsen P, Vilstrup H, Andersen PK. The clinical course of cirrhosis: The importance of multistate models and competing risks analysis. Hepatology. 2015 Jul 1;62(1):292–302.
- 17. Clinical Practice Research Datalink [Internet]. 2012. Available from: http://www.cprd.com/intro.asp
- 18. NHS Information Centre. Hospital Episodes Statistics [Internet]. Health & Social Care Information Centre. 2015. Available from: http://www.hscic.gov.uk/hes
- 19. Lynge E, Sandegaard JL, Rebolj M. The Danish National Patient Register. Scand J Public Health. 2011 Jul 1;39(7 suppl):30–3.
- 20. Pedersen CB. The Danish Civil Registration System. Scand J Public Health. 2011 Jul 1;39(7_suppl):22–5.
- 21. Chisholm J. The Read clinical classification. BMJ. 1990 Apr 28;300(6732).
- 22. Ratib S, West J, Crooks CJ, Fleming KM. Diagnosis of Liver Cirrhosis in England, a Cohort Study, 1998-2009: A Comparison with Cancer. Am J Gastroenterol. 2014 Feb;109(2):190–8.
- 23. Fleming KM, Aithal GP, Solaymani-Dodaran M, Card TR, West J. Incidence and prevalence of cirrhosis in the United Kingdom, 1992–2001: A general population-based study. J Hepatol. 11;49(5):732–8.
- 24. Van Staa TP, Abenhaim L, Cooper C, Zhang B, Leufkens HGM. The use of a large pharmacoepidemiological database to study exposure to oral corticosteroids and risk of fractures: validation of study population and results. Pharmacoepidemiol Drug Saf. 2000 Sep 1;9(5):359–66.
- 25. de Franchis R. Evolving Consensus in Portal Hypertension Report of the Baveno IV Consensus Workshop on methodology of diagnosis and therapy in portal hypertension. J Hepatol. 2005 Jul 1;43(1):167–76.
- Charlson ME, Pompei P, Ales KL, MacKenzie CR. A new method of classifying prognostic comorbidity in longitudinal studies: development and validation. J Chronic Dis. 1987;40(5):373–83.
- 27. Panday K, Gona A, Humphrey MB. Medication-induced osteoporosis: screening and treatment strategies. Ther Adv Musculoskelet Dis. 2014 Oct;6(5):185–202.
- 28. McCaughan GW, Feller RB. Osteoporosis in Chronic Liver Disease: Pathogenesis, Risk Factors, and Management. Dig Dis. 1994;12(4):223–31.
- 29. Luxon BA. Bone Disorders in Chronic Liver Diseases. Curr Gastroenterol Rep. 2011;13(1):40–8.

- Vestberg K, Thulstrup AM, Sørensen HT, Ottesen P, Sabroe S, Vilstrup H. Data Quality of Administratively Collected Hospital Discharge Data for Liver Cirrhosis Epidemiology. J Med Syst. 1997;21(1):11–20.
- 31. Van Staa TP, Leufkens HGM, Abenhaim L, Zhang B, Cooper C. Use of Oral Corticosteroids and Risk of Fractures. J Bone Miner Res. 2000 Jun 1;15(6):993–1000.
- 32. Schmidt M, Schmidt SAJ, Sandegaard JL, Ehrenstein V, Pedersen L, Sørensen HT. The Danish National Patient Registry: a review of content, data quality, and research potential. Clin Epidemiol. 2015;7:449–90.
- 33. Fleming KM, Aithal GP, Card TR, West J. All-cause mortality in people with cirrhosis compared with the general population: a population-based cohort study. Liver Int. 2012 Jan 1;32(1):79–84.
- 34. Ninkovic M, Skingle SJ, Bearcroft PW, Bishop N, Alexander GJ, Compston JE. Incidence of vertebral fractures in the first three months after orthotopic liver transplantation. Eur J Gastroenterol Hepatol. 2000 Aug;12(8):931–5.
- 35. Diamond T, Stiel D, Lunzer M, Wilkinson M, Roche J, Posen S. Osteoporosis and skeletal fractures in chronic liver disease. Gut. 1990 Jan;31(1):82–7.
- Carey EJ, Balan V, Kremers WK, Hay JE. Osteopenia and osteoporosis in patients with end-stage liver disease caused by hepatitis C and alcoholic liver disease: not just a cholestatic problem. Liver Transplant Off Publ Am Assoc Study Liver Dis Int Liver Transplant Soc. 2003 Nov;9(11):1166–73.
- 37. Lindsell D, Wilson A, Maxwell J. Fractures on the chest radiograph in detection of alcoholic liver disease. Br Med J Clin Res Ed. 1982;285(6342):597–9.
- Gonzalez-Reimers E, Alvisa-Negrin J, Santolaria-Fernandez F, Candelaria Martin-Gonzalez M, Hernandez-Betancor I, Fernandez-Rodriguez CM, et al. Vitamin D and nutritional status are related to bone fractures in alcoholics. Alcohol Alcohol Oxf Oxfs. 2011 Apr;46(2):148–55.
- Wibaux C, Legroux-Gerot I, Dharancy S, Boleslawski E, Declerck N, Canva V, et al. Assessing bone status in patients awaiting liver transplantation. Jt Bone Spine Rev Rhum. 2011 Jul;78(4):387–91.
- 40. Tsai C-F, Liu C-J, Chen T-J, Chu C-J, Lin H-C, Lee F-Y, et al. Increased incidence of orthopedic fractures in cirrhotic patients: A nationwide population-based study. J Hepatol. 2013 Apr;58(4):706–14.
- Guañabens N, Cerdá D, Monegal A, Pons F, Caballería L, Peris P, et al. Low Bone Mass and Severity of Cholestasis Affect Fracture Risk in Patients With Primary Biliary Cirrhosis. Gastroenterology. 2010 Jun;138(7):2348–56.
- 42. Davila JA, Morgan RO, Richardson PA, Du XL, McGlynn KA, El-Serag HB. Use of surveillance for hepatocellular carcinoma among patients with cirrhosis in the United States. Hepatology. 2010 Jul 1;52(1):132–41.
- 43. Sangiovanni A, Colombo M. Surveillance for hepatocellular carcinoma: A standard of care, not a clinical option. Hepatology. 2011 Dec 1;54(6):1898–900.

20

Downloaded for Anonymous User (n/a) at Aalborg University Hospital from ClinicalKey.com by Elsevier on April 23, 2018. For personal use only. No other uses without permission. Copyright ©2018. Elsevier Inc. All rights reserved.

- 44. Jepsen P, Ott P, Andersen P, Sørensen H, Vilstrup H. Risk for hepatocellular carcinoma in patients with alcoholic cirrhosis: A danish nationwide cohort study. Ann Intern Med. 2012 Jun 19;156(12):841–7.
- Tzortziou Brown V, Underwood M, Mohamed N, Westwood O, Morrissey D. Professional interventions for general practitioners on the management of musculoskeletal conditions. Cochrane Database Syst Rev. 2016 May 6;(5):CD007495.
- Byun J-H, Jang S, Lee S, Park S, Yoon HK, Yoon B-H, et al. The Efficacy of Bisphosphonates for Prevention of Osteoporotic Fracture: An Update Meta-analysis. J Bone Metab. 2017 Feb;24(1):37–49.
- Compston J, Cooper A, Cooper C, Gittoes N, Gregson C, Harvey N, et al. UK clinical guideline for the prevention and treatment of osteoporosis. Arch Osteoporos. 2017;12(1):43.
- Avenell A, Mak JC, O'Connell D. Vitamin D and vitamin D analogues for preventing fractures in post-menopausal women and older men. Cochrane Database Syst Rev [Internet]. 2014;(4). Available from: http://dx.doi.org/10.1002/14651858.CD000227.pub4
- 49. Lambrinoudaki I, Christodoulakos G, Botsis D. Bisphosphonates. Ann N Y Acad Sci. 2006 Dec 1;1092(1):397–402.
- Bischoff-Ferrari HA, Willett WC, Wong JB, Giovannucci E, Dietrich T, Dawson-Hughes B. Fracture prevention with vitamin d supplementation: A meta-analysis of randomized controlled trials. JAMA. 2005 May 11;293(18):2257–64.
- El-Khoury F, Cassou B, Charles M-A, Dargent-Molina P. The effect of fall prevention exercise programmes on fall induced injuries in community dwelling older adults: systematic review and meta-analysis of randomised controlled trials. BMJ [Internet]. 2013 Oct 29;347. Available from: http://www.bmj.com/content/347/bmj.f6234.abstract

CCE

	UK		Denmark			
	Cirrhosis patients N=3,706	Controls N=36,854	Cirrhosis patients N=17,779	Controls N=80,815		
Age at diagnosis (%)						
≤44	645 (17.4)	7,426 (20.1)	2,117 (11.9)	10,033 (12.4)		
45–54	1,089(29.4)	10,227(27.8)	5,379 (30.3)	24,950 (30.9)		
55–64	1,109 (29.9)	10,439(28.3)	6,308 (35.5)	28,604 (35.4)		
65–74	623 (16.8)	6,181 (16.8)	3,221 (18.1)	14,113 (18.1)		
≥75	240 (6.5)	2,581 (7.0)	754 (4.2)	3,115 (3.9)		
Median(IQR) age at diagnosis (years)	56 (47-64)	55(47-64)	57 (50–64)	57 (50–64)		
Sex (%)						
Males	2,520 (68.0)	25,074 (68.0)	11,687 (66.8)	52,633 (65.1)		
Female	1,186 (32.0)	11,780 (32.0)	5,912 (33.3)	28,182 (34.9)		
Smoking status						
Smoker	1,673(45.1)	7,638(20.7)				
Ex-smoker	647(17.5)	6,589(17.9)				
Never smoked	791(21.3)	13,231(35.9)				
No available data	595(16.1)	9,396(25.5)				
Disease stage ^a						
Compensated	1,859 (50.2)	-	12,484 (70.2)			
Decompensated	1,847 (49.8)	-	5,295 (29.8)			
Charlson comorbidity score (%)						
0	2,310 (62.3)	34,262 (92.7)	11,276 (63.4)	76,403 (94.5)		
1	793 (21.4)	1,358 (3.7)	3,043 (17.1)	2,347 (2.9)		
2	603 (16.3)	1,234 (3.4)	3,460 (19.5)	2,065 (2.6)		
Evidence of osteoporosis						
Diagnostic code +/- bisphosphonate	234(6.3)	1133(3.1)	547 (3.1)	1,366 (1.7)		
prescription						
Previous densitometry scan	<u>3(0.01)</u>	<u>6(0.16)</u>	16(0.09)	<u>30(0.04)</u>		
Other cirrhosis aetiology ^b						
Autoimmune disease	30(0.8)	-	44 (0.25)	-		
Metabolic disease	157(4.2)	-	303 (1.7)	-		
Viral hepatitis	243(6.6)	-	627 (3.5)	-		
Only alcohol-related	3,290(88.8)	-	16,824 (94.6)	-		

Table 1. Demographic characteristics of alcoholic cirrhosis cohort and population controls in Denmark and England.

^aBaveno classification ^bpatients can have more than one other aetiology in addition to alcohol

Table 2: Hip fracture rates and hazard ratios (HR) for alcoholic cirrhosis patients and population controls in the Danish and UK sample according to subgroups defined by age, gender, and decompensation status with 95% confidence intervals

	English Cirrhosis patients				Danish Cirrhosis patients			
	No. of hip fracture events	Person years	Rate per 1,000 person years	Adjusted HR ¹ (95% Cl)	No. of hip fracture events	Person- years	Rate per 1,000 person years	Adjusted HR ¹ (95% Cl)
Overall	108	9,483	11.4 [9.4 - 13.8]	5.5[4.3 - 6.9]	977	61,122	16.0[15.0, 17.0]	8.5[7.8–9.3]
Min-44	14	1,822	7.7 [4.6 - 12.9]	17.9 [7.8 - 40.8]	77	10,546	7.3 [5.8–9.1]	16.4 [11.4–23.6]
45–54	31	3,099	10.0 [7.0 - 14.2]	14.7[8.6 - 24.9]	297	21,712	13.7 [12.2–15.3]	13.1 [10.9–15.7]
55-64	33	2,958	11.2[7.9 - 15.7]	6.2[3.9 - 9.5]	363	20,360	17.8 [16.0–19.8]	9.8 [8.4–11.4]
65-74	19	1,226	15.5 [9.9 - 24.3]	3.2[1.9 - 5.4]	193	7361	26.2 [22.7–30.2]	5.9 [4.9–7.1]
75+	11	378	29.1[16.1 - 52.6]	2.1[1.1 - 2.2]	47	1,143	41.1 [30.2–54.7]	3.0 [2.1–4.2]
Men	59	6,259	9.4 [7.3 - 12.2]	4.4[3.2 - 6.1]	571	38,638	14.8 [13.7–16.0]	9.0 [8.0–10.1]
Women	49	3,224	15.2 [11.5 - 20.1]	7.1[5.0 - 9.9]	406	22,484	18.1 [16.3–19.9]	7.9 [6.9–9.0]
Compensated	53	5,621	9.4 [7.2 -12.3]	-	665	42,480	15.7 [14.5–16.9]	-
Decompensated	55	3,862	14.2 [10.9 -18.5]	-	312	18,642	16.7 [14.9–18.7]	-
Decompensated and/or HE ²	61	4,367	13.9[10.9-17.9]		418	24,382	17.1 [15.5–18.9]	-
Evidence of osteoporosis	26	837	31.0[21.1-45.6]	-	44	1,245	35.3 [25.7–47.4]	-
None	82	8,646	9.5[7.6-11.8]	-	503	59,877	15.6 [14.6–16.6]	-

¹HR adjusted for age, sex and comorbidity for cirrhosis patients vs. population controls

² HE-Hepatic Encephalopathy, (includes alcoholic encephalopathy and Wernicke's syndrome)

Table 3: Cirrhosis patients and population controls' 5-year risk of a hip fracture by age group at inclusion, gender and compensation status with 95% confidence intervals (Cl's)

	En	ngland	De		
	Cirrhosis patients	Population controls	Cirrhosis patients	Population contro	ls
Overall	2.9[2.3-3.6]	0.8[0.7-0.9]	4.6 (4.3–5.0]	0.9 [0.8–0.9]	
Age group					Table 4:
Min-44	2.2 [0.9 - 4.4]	0.2 [0.1 - 0.5]	2.5 [1.9–3.3]	0.1 [0.1–0.2]	Alcohol
45–54	2.3 [1.4 - 3.4]	0.4 [0.3 - 0.6]	4.3 [3.8–4.9]	0.4 [0.3–0.5]	ic
55–64	2.7 [1.8 - 3.9]	0.7 [0.6 - 0.9]	5.0 [4.4–5.6]	0.7 [0.6–0.8]	cirrhosi
65–74	2.0 [1.1 - 3.2]	1.2 [1.1 - 3.1]	5.9 [5.1–6.8]	1.6 [1.4–1.6]	S
75–	3.8 [1.9 - 6.7]	2.6 [2.1 - 3.1]	5.9 [4.3–8.0]	5.5 [4.7–6.5]	patient
Sex			~		s and populat
Male	2.4 [1.8 - 3.3]	0.8 [0.7 - 1.0]	4.2 [3.9–4.6]	0.7 [0.7–0.8]	ion
Female	4.3 [3.1 - 5.8]	1.2 [1.0 - 1.6]	5.5 [4.9–6.1]	1.1 [1.0–1.2]	control
Disease severity					s' 30- day
Compensated	3.1 [2.3 - 4.1]	-	4.7 [4.3–5.1]	-	mortali
Decompensated	3.1 [2.3 - 4.2]		4.6 [4.0–5.2]	-	ty risk (%) after a

hip fracture with 95% confidence intervals (CI's)

24





(a) English data



Figure 2: 5-year cumulative incidence for hip fractures by age group at inclusion.

Downloaded for Anonymous User (n/a) at Aalborg University Hospital from ClinicalKey.com by Elsevier on April 23, 2018. For personal use only. No other uses without permission. Copyright ©2018. Elsevier Inc. All rights reserved.



5-year cumulative incidence for hip fractures by age group at inclusion.



Highlights

- Hip fracture risk is raised by at least 5 fold in an alcoholic cirrhosis population compared to population controls of similar age and sex
- The increased risks of hip fracture is higher in younger adults
- The cumulative incidence of hip fracture (having taken account of the competing risk of death) is greater than that of hepatocellular carcinoma (HCC) in this population.